Dear Editor-in-Chief Lian-Sheng Ma,

Please find enclosed the revised manuscript entitled “Pediatric vascular tumors of the liver: review from the pathologist’s point of view”.

We would like to thank the reviewers for their valuable comments, which have improved our manuscript.

With this revised manuscript, we accommodated the remarks of the reviewers and editorial office (64884). Please find our answers to his/her comments below. All changes in the revised manuscript are marked in red.

The infantile hepatic hemangioendothelioma (IHHE) presented in many clinical studies in the literature was not clearly presented in the review with its synonymous terminology from the pathologist’s point of view. Please summarize the classification of presented hepatic vascular tumors presented in the review in either a table or a figure.

An overview table is now added to the revised manuscript (see Table 1).

Please spell out any used abbreviation on its first appearance.

The text has been searched for abbreviations and all abbreviations are spelled out on its first appearance.

Reference 1 is not the suitable one. Correct references of citations.

References are updated in the revised manuscript.

Clarify why the authors focused on vascular tumor of liver. Clarify how's important of liver vascular tumor.

This review is an invited manuscript. This title was proposed to and approved by the editorial board.
The manuscript should be corrected for grammatical and spelling errors.

The text has been searched and corrected for grammatical and spelling errors.

The “Author Contributions” section is missing.

The author contribution is added in the revised manuscript.

The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.

All figures and tables are originals and have not been published elsewhere. The figures are arranged using PowerPoint.

We hope that our responses are satisfying to the reviewers and the editorial office, and that our contribution will now be considered for publication in *World Journal of Hepatology*.

Kind regards,

Dr. Fleur Cordier, MD  
Prof. Dr. David Creytens, MD, PhD  
Department of Pathology  
Ghent University  
Ghent University Hospital, PAD, Building 23  
De Pintelaan 185  
B-9000 Ghent  
Belgium  
e-mail: fleur.cordier@uzgent.be, david.creytens@uzgent.be